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1008 の作業状況

2024 2024

Today

Jan Feb Mar

TPC HV testing Jan 30 - Feb 6

TPC cosmic ray run Feb 6

TPC laser bench installation Feb 7 - Feb 11

Final TPC test Feb 12

SOB/MJACK Feb 5 - Mar 1

Staging ROC's for installation Feb 8 - Feb 12

Install and test South ROC Feb 13 - Feb 26

Install and test North ROC Feb 21 - Mar 1

Install X wing Feb 26 - Mar 1

Install MVTX Feb 28 - Mar 8

installation_2024-02-01

sPHENIX Re-Installation

sPHENIX Planning meeting (2/1)

Current Work

2

• Prepared and installed HV cables.

• Capacitance test of HV cable complete.

• North and south FEE’s are in.

• Finished installing resistors in the contingency boxes.

• Working on SOP for canary chamber.

• Continued work on lasers.  Parts are here and assembly is ongoing.  Shooting for first laser 
installation on 2/7.

• 今日（2024/02/01）1008 の作業計画について報
告があった


• TPC の作業が遅れており、全体的に 1 週間計画
がずれそう


• INTT の作業は北側 ROC インストールの時間がか
なり削られ、厳しい戦いになる

平日 11 日間、休日 4 日
平日 8 日間、休日 2 日

https://indico.bnl.gov/event/22117/


ラックルームでの FELIX を使ったキャリブレーション
• ROC 再インストールの前に FELIX でキャリブレーションが取れることを確認し
たい。


• ラウールがある程度作業して・・・いなかった


• セットアップ構築中。ファイバー接続完了もクロック信号見えず。 
パッチパネルとファイバーのバージョンが合っていない。 
明日、シリコンラボからパッチパネルを移設する。


• キャリブレーションに必要なスケジューラが無い。測定不可能では？

2024/02/01 作業開始時 2024/02/01 作業中

2024/02/01 作業中

パッチパネル裏側

追加

追加

取り外した


